

Inappropriate antibiotic prescribing and its determinants among outpatient children in 3 low- and middle-income countries: A multicentric community-based cohort study

Antoine Ardillon, Lison Ramblière, Elsa Kermorvant-Duchemin, Touch Sok, Andrianirina Zafitsara Zo, Jean-Baptiste Diouf, Pring Long, Siyin Lach, Fatoumata Diene Sarr, Laurence Borand, et al.

▶ To cite this version:

Antoine Ardillon, Lison Ramblière, Elsa Kermorvant-Duchemin, Touch Sok, Andrianirina Zafitsara Zo, et al.. Inappropriate antibiotic prescribing and its determinants among outpatient children in 3 low- and middle-income countries: A multicentric community-based cohort study. PLoS Medicine, 2023, 20 (6), pp.e1004211. 10.1371/journal.pmed.1004211. pasteur-04122106

HAL Id: pasteur-04122106 https://pasteur.hal.science/pasteur-04122106v1

Submitted on 8 Jun2023

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution 4.0 International License



Citation: Ardillon A, Ramblière L, Kermorvant-Duchemin E, Sok T, Zo AZ, Diouf J-B, et al. (2023) Inappropriate antibiotic prescribing and its determinants among outpatient children in 3 lowand middle-income countries: A multicentric community-based cohort study. PLoS Med 20(6): e1004211. https://doi.org/10.1371/journal. pmed.1004211

Received: September 22, 2022

Accepted: March 3, 2023

Published: June 6, 2023

Copyright: © 2023 Ardillon et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Data cannot be shared publicly because data privacy is subject to French, Cambodian, Senegalese and Malagasy regulations. Data are available from the Ethics Committee (contact via irb@pasteur.fr) for researchers who meet the criteria for access to confidential data.

Funding: This work was supported by the Department of International Cooperation of the Principality of Monaco (https://en.gouv.mc/ Government-Institutions/The-Government/ Ministry-of-Foreign-Affairs-and-Cooperation/ RESEARCH ARTICLE

Inappropriate antibiotic prescribing and its determinants among outpatient children in 3 low- and middle-income countries: A multicentric community-based cohort study

Antoine Ardillon ^{1,2°}, Lison Ramblière ^{1,2°}, Elsa Kermorvant-Duchemin ³, Touch Sok ⁴, Andrianirina Zafitsara Zo⁵, Jean-Baptiste Diouf⁶, Pring Long⁷, Siyin Lach⁷, Fatoumata Diene Sarr⁸, Laurence Borand^{7,8}, Felix Cheysson ⁹, Jean-Marc Collard ^{10°}, Perlinot Herindrainy ^{11°}, Agathe de Lauzanne^{7°}, Muriel Vray^{12°}, Elisabeth Delarocque-Astagneau ^{1,13}, Didier Guillemot^{1,2,13}, Bich-Tram Huynh ^{1,2*}, On behalf of the BIRDY study group¹¹

 Université Paris-Saclay, UVSQ, Inserm, CESP, Anti-infective evasion and pharmacoepidemiology team, Montigny-Le-Bretonneux, France, 2 Institut Pasteur, Université Paris Cité, Epidemiology and Modelling of Antibiotic Evasion (EMAE), Paris, France, 3 AP-HP, Hôpital Necker-Enfants malades, Department of Neonatal medicine, Université Paris Cité, Paris, France, 4 Ministry of Health, Phnom Penh, Cambodia,
Peadiatric Ward, Centre Hospitalier de Soavinandriana, Antananarivo, Madagascar, 6 Centre Hospitalier Roi Baudouin Guédiawaye, Dakar, Sénégal, 7 Epidemiology and Public Health Unit, Institut Pasteur du Cambodge, Phnom Penh, Cambodia, 8 Center for Tuberculosis Research, Division of Infectious Diseases, Johns Hopkins University School of Medicine, Baltimore, Maryland, United States of America, 9 Sorbonne Université, UMR CNRS 8001, LPSM, Paris, France, 10 Experimental Bacteriology Unit, Institut Pasteur de Madagascar, Antananarivo, Madagascar, 11 Epidemiology Unit, Institut Pasteur de Dakar, Dakar, Sénégal, 13 AP-HP. Paris Saclay, Public Health, Medical Information, Clinical research, Le Kremlin-Bicêtre, France

• These authors contributed equally to this work.

¶ Membership of the BIRDY study group is provided in the Acknowledgements. * bich-tram.huynh@pasteur.fr

Abstract

Background

Antibiotic resistance is a global public health issue, particularly in low- and middle-income countries (LMICs), where antibiotics required to treat resistant infections are not affordable. LMICs also bear a disproportionately high burden of bacterial diseases, particularly among children, and resistance jeopardizes progress made in these areas. Although outpatient antibiotic use is a major driver of antibiotic resistance, data on inappropriate antibiotic prescribing in LMICs are scarce at the community level, where the majority of prescribing occurs. Here, we aimed to characterize inappropriate antibiotic prescribing among young outpatient children and to identify its determinants in 3 LMICs.

Methods and findings

We used data from a prospective, community-based mother-and-child cohort (BIRDY, 2012 to 2018) conducted across urban and rural sites in Madagascar, Senegal, and

Department-of-International-Cooperation), MSDAVENIR (http://www.msdavenir.fr/) and Total foundation (https://www.foundation.total/fr). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Abbreviations: AIC, Akaike information criterion; aOR, adjusted odds ratio; CI, confidence interval; HIC, high-income country; IMCI, Integrated Management of Childhood Illness; LMIC, low- and middle-income country; WHO, World Health Organization. Cambodia. Children were included at birth and followed-up for 3 to 24 months. Data from all outpatient consultations and antibiotics prescriptions were recorded. We defined inappropriate prescriptions as antibiotics prescribed for a health event determined not to require antibiotic therapy (antibiotic duration, dosage, and formulation were not considered). Antibiotic appropriateness was determined a posteriori using a classification algorithm developed according to international clinical guidelines. We used mixed logistic analyses to investigate risk factors for antibiotic prescription during consultations in which children were determined not to require antibiotics. Among the 2,719 children included in this analysis, there were 11,762 outpatient consultations over the follow-up period, of which 3,448 resulted in antibiotic prescription. Overall, 76.5% of consultations resulting in antibiotic prescription were determined not to require antibiotics, ranging from 71.5% in Madagascar to 83.3% in Cambodia. Among the 10,416 consultations (88.6%) determined not to require antibiotic therapy, 25.3% (n = 2,639) nonetheless resulted in antibiotic prescription. This proportion was much lower in Madagascar (15.6%) than in Cambodia (57.0%) or Senegal (57.2%) (p < 0.001). Among the consultations determined not to require antibiotics, in both Cambodia and Madagascar the diagnoses accounting for the greatest absolute share of inappropriate prescribing were rhinopharyngitis (59.0% of associated consultations in Cambodia, 7.9% in Madagascar) and gastroenteritis without evidence of blood in the stool (61.6% and 24.6%, respectively). In Senegal, uncomplicated bronchiolitis accounted for the greatest number of inappropriate prescriptions (84.4% of associated consultations). Across all inappropriate prescriptions, the most frequently prescribed antibiotic was amoxicillin in Cambodia and Madagascar (42.1% and 29.2%, respectively) and cefixime in Senegal (31.2%). Covariates associated with an increased risk of inappropriate prescription include patient age greater than 3 months (adjusted odds ratios (aOR) with 95% confidence interval (95% CI) ranged across countries from 1.91 [1.63, 2.25] to 5.25 [3.85, 7.15], p < 0.001) and living in rural as opposed to urban settings (aOR ranged across countries from 1.83 [1.57, 2.14] to 4.40 [2.34, 8.28], p < 0.001). Diagnosis with a higher severity score was also associated with an increased risk of inappropriate prescription (aOR = 2.00 [1.75, 2.30] for moderately severe, 3.10 [2.47, 3.91] for most severe, p < 0.001), as was consultation during the rainy season (aOR = 1.32 [1.19, 1.47], p < 0.001). The main limitation of our study is the lack of bacteriological documentation, which may have resulted in some diagnosis misclassification and possible overestimation of inappropriate antibiotic prescription.

Conclusion

In this study, we observed extensive inappropriate antibiotic prescribing among pediatric outpatients in Madagascar, Senegal, and Cambodia. Despite great intercountry heterogeneity in prescribing practices, we identified common risk factors for inappropriate prescription. This underscores the importance of implementing local programs to optimize antibiotic prescribing at the community level in LMICs.

Author summary

Why was this study done?

- Antibiotic resistance is a major public health issue in low- and middle-income countries (LMICs), particularly among children, who face a disproportionately large share of the global burden of bacterial disease.
- Antibiotic overuse, as a result of inappropriate prescribing, is a major driver of antibiotic resistance, but relevant data from LMICs are scarce at the community level, where the majority of antibiotic prescribing occurs.
- Data regarding the prevalence and risk factors of inappropriate prescribing among outpatient children in LMICs are needed to understand its extent and causes and to inform intervention.

What did the researchers do and find?

- Using data from a mother-and-child cohort in urban and rural sites of Madagascar, Senegal, and Cambodia, the appropriateness of outpatient antibiotic prescribing was assessed among all children (N = 2,719) having at least 1 outpatient consultation from birth up to 24 months of age.
- We identified that approximately 3 quarters of all antibiotic prescriptions were inappropriate.
- We found that 15.5%, 57.0%, and 57.2% of consultations not requiring antibiotic therapy, nonetheless, resulted in antibiotic prescription in Madagascar, Cambodia, and Senegal, respectively.
- Risk factors for inappropriate prescription include older patient age (being older than 3 months), disease with a higher severity score, consultation during the rainy season, and residence in rural areas.

What do these findings mean?

- Inappropriate antibiotic prescribing is pervasive among outpatient children in LMICs, although there is great heterogeneity across countries in prescribing practices.
- Locally adapted antibiotic stewardship programs are needed to optimize pediatric antibiotic prescribing, while taking into account the specific contexts associated with highrisk populations across diverse LMICs.

Introduction

Antibiotic resistance is a leading global threat to public health [1]. Over the past 2 decades, there has been an alarming increase in the number of pathogenic bacteria that have developed resistance to common antibacterial agents, particularly in low- and middle-income countries (LMICs), where antibiotic-resistant bacteria appear to be more prevalent than in high-income countries (HICs) [2]. Recent increases in antibiotic resistance in LMICs may be explained by

various factors, including changing prescribing patterns among healthcare providers, and increasing patient access both to formal healthcare services and to informal or "over-the-counter" antibiotics. Together, these trends have translated to increases in overall antibiotic consumption in the general community in LMICs over recent years [3,4].

Inappropriate antibiotic prescribing is a primary factor underlying the emergence of antibiotic resistance. An antibiotic prescription may be deemed inappropriate if the chosen dose, molecule, or schedule is nonoptimal, or if the patient's clinical indication does not necessitate antibiotic therapy. In this study, we focus on the latter. Inappropriate antibiotic prescribing has deleterious impacts on human health across scales. At the individual level, patients are unnecessarily exposed to potential side effects of antibiotic consumption, including allergic reactions and digestive symptoms related to microbiome dysbiosis, and may experience delays to appropriate care. Potential longer-term risks of pediatric antibiotic exposure, such as increased risks of asthma and autoimmune disorders, continue to be described [5]. At the population level, inappropriate antibiotic prescribing creates additional selection pressure favoring the emergence and spread of bacterial strains and serotypes that resist the desired clinical effects of antibiotics [6,7].

In the context of efforts to mitigate the burden of antibiotic resistance, reducing inappropriate antibiotic prescribing is a public health priority. However, the share of antibiotic prescribing that qualifies as inappropriate is poorly described in LMICs. Most studies estimating the magnitude of and risk factors for inappropriate antibiotic prescribing have been based in HICs [8–11]. In particular, there are few studies conducted in LMICs among outpatient children even though a large share of antibiotic prescribing occurs in primary care, and young children in LMICs bear the greatest share of global infectious disease burden and are hence heavily exposed to antibiotics in the first years of life [12,13]. Yet data describing the magnitude and determinants of inappropriate antibiotic prescribing among children in LMICs are lacking. Such data are necessary to quantify potential for prescribing reductions, to describe which populations are potentially overexposed to antibiotics, and ultimately to inform interventions targeting prevention of inappropriate prescribing.

The objectives of this study were to describe and quantify inappropriate antibiotic prescribing and to identify determinants of inappropriate prescribing among children enrolled in a community-based cohort in Madagascar, Cambodia, and Senegal.

Methods

Study design

We used data from the BIRDY cohort (Bacterial Infections and antibiotic Resistant Diseases among Young children in LMICs), a prospective, multicentric, community-based motherand-child cohort. The study took place across urban and rural sites in Madagascar (pilot phase: 2012 to 2014 and complete phase: 2014 to 2018), Senegal (2014 to 2018), and Cambodia (2014 to 2018). The primary objective of this cohort study was to estimate the incidence of antibiotic-resistant bacterial infections in children.

The formal analysis plan for the BIRDY cohort was established in 2012 [14]. There were no changes in the protocol between the pilot phase and complete phase that may bias results of the present study. No additional formal prospective analysis plan specific to this study on inappropriate antibiotic prescribing was made. All methodological details, including development of the classification algorithm for antibiotic appropriateness, definition of outcomes and covariates and statistical methodology, were discussed prior to data analysis in 2021. There were no modifications to this initial planned analysis or main hypotheses subsequent to data exploration. Changes considered subsequent to peer review are described below where

appropriate. This study is reported as per the Strengthening the Reporting of Observational Studies in Epidemiology (STROBE) guideline (<u>S1 Checklist</u>).

This BIRDY cohort has been described in detail elsewhere [14]. Briefly, in Cambodia, the study was conducted in 2 urban districts of Phnom Penh (population: 87,035) and 2 rural districts of Kampong Speu (population: 79,000). In Madagascar, the study was conducted in 3 urban districts of Antananarivo (population: 14,997) and in the semirural city of Moramanga (population: 17,159). In Senegal, the study was conducted in an urban district of Guédiawaye (population: 20,529) and in the rural community of Sokone (population: 14,500).

To recruit study participants, a first phase of exhaustive identification of all pregnant women was carried out in the study areas. Women were enrolled, after having given written informed consent, during their third trimester of pregnancy and followed until delivery to include their newborns at birth.

Inclusion criteria were the following for both the mother and her newborn: routine residence in the study area with no plan to move away during the follow-up period, and no opposition to the research being conducted, nor to the collection of epidemiological data and biological samples. Data regarding sociodemographic, medical, and obstetric characteristics of mothers were collected, as well as data regarding the delivery and health characteristics of newborns.

After delivery, children were followed up for between 3 and 24 months, depending on the site and country. Duration of follow-up was 6 months during the pilot phase and 18 months during the complete phase in Madagascar and 24 months in Cambodia. In Senegal, children were followed up for 3 months in rural areas and 12 months in urban areas. Children were followed up by both active and passive surveillance according to the same schedule across all study sites. Active surveillance consisted of scheduled home visits within 3 days of childbirth, followed by weekly visits until the first month of life, fortnightly visits from the first to the third months of life, and finally, monthly visits until the end of the study period. During these visits, children's health status was checked, especially for signs of infection. In case of a health event, the child was referred to an outpatient consultation or directly to hospital if the child's condition was serious. Passive surveillance consisted of asking mothers to contact the investigation team whenever the child showed signs of infection in order to schedule an outpatient consultation. Thus, children were able to benefit from outpatient consultations whenever they were sick, whether identified by their guardians or by study investigators. Clinical decisions regarding antibiotic prescribing were not monitored or imposed, but were rather left to the attending physicians to decide according to local protocols.

Importantly, epidemiological data collected during these consultations included children's diagnoses, symptoms, and treatments received, including details of any antibiotics prescribed. Costs of consultation and antimicrobial treatment were covered by the study.

Study population and outcomes

For the present study, the population considered was all children from the BIRDY cohort who had at least 1 outpatient consultation during their follow-up. Only systemic antibiotics were taken into account (oral, intramuscular, and intravenous). We did not consider local antibiotics (dermal and ophthalmic) because their impact on antibiotic resistance is probably different from systemic antibiotics.

For each consultation, diagnoses were categorized by the study team using data collected during children consultation. First, we ensured that all diagnoses made by the physician not of infectious origin were concordant with data collected on the signs and symptoms presented by the child. Then, all infectious diagnoses were determined using a syndromic algorithm based on the World Health Organization (WHO) Integrated Management of Childhood Illness (IMCI) guidelines. This algorithm was developed previously and is described elsewhere [15]. All health events, whether infectious or not, that were classified as "uncertain" or not classified by the algorithm were reviewed by a pediatrician (EK) and a medical epidemiologist (BTH).

We defined inappropriate antibiotic prescriptions as those being prescribed during an outpatient consultation for a child's health event that does not require antibiotic therapy, regardless of the duration of prescription, type of antibiotic, and dosage.

We constructed a standardized classification algorithm in 2 steps to determine which health events require antibiotics or not (S1 Table). The first step was to define diagnoses that do not require antibiotic therapy. For some diagnoses, the indication for antibiotic therapy depends on other discriminating criteria, such as age and/or signs of severity and/or associated symptoms. Thus, the second step was to define criteria to classify the remaining health events that could not be classified solely based upon diagnosis in the first step. These criteria were defined based on the IMCI guidelines and the French pediatric reference manual "Pédiatrie pour le Praticien–médecine de l'enfant et de l'adolescent" [16,17].

Some children received multiple diagnoses and/or multiple antibiotic prescriptions during a single consultation; hence, the number of consultations not requiring antibiotics and the number of antibiotics prescribed among consultations not requiring antibiotics are considered as distinct outcomes. For children receiving multiple diagnoses during their consultation, antibiotic therapy was defined as inappropriate provided that none of the diagnoses required antibiotic therapy. For children receiving prescriptions for multiple antibiotic classes or molecules during a single consultation, all prescriptions were defined as inappropriate provided that none of the corresponding clinical diagnoses required antibiotic therapy.

Statistical analyses

First, we calculated basic descriptive statistics of characteristics of the study population in each country, including the total number of consultations, diagnoses, antibiotic prescriptions, and inappropriate antibiotic prescriptions. Comparisons between countries were made using the chi2 test or exact Fisher test (depending on the variable's distribution) for qualitative variables and the Student test or nonparametric Mann–Whitney test (depending on the variable's distribution) for quantitative variables. All tests were bilateral, and the significance threshold was set at 0.05.

Second, we evaluated risk factors for inappropriate antibiotic prescribing. Specifically, we investigated whether antibiotic prescription during outpatient consultations among children not requiring antibiotic therapy was associated with exposures of primary interest, defined as characteristics of the health event (history of the health event, severity score, season, and child's age). We considered site, child sex, child nutritional status, and sociodemographic, maternal, and pregnancy characteristics as potential confounders. We included all consultations among children whose health event was determined not to require antibiotic therapy by the classification algorithm described above.

Nutritional status was defined using the z-score, adjusted according to the child's age and weight using WHO reference tables, where children with a z-score < -2 were considered underweight [18]. When information on the child's weight at consultation was missing, weight z-score at the closest consultation or visit was used. The severity of each health event was characterized based on clinical signs observed by the physician and according to the WHO IMCI standard [16]. A clinical severity score variable was defined as an ordinal variable with 3 values, ranging from 0 (least severe) to 2 (most severe). House density was defined as the number of people living in the household divided by the number of sleeping rooms in the household.

Overcrowding was defined as a house density ≥ 4 , corresponding to the third quartile of the house density variable. We considered the season in which the consultation occurred, defined as rainy versus not. The rainy season was defined as April to September in Cambodia, November to April in Madagascar, and July to November in Senegal. We considered a delivery to be complicated if the child was born by cesarean section, if the child required resuscitation following delivery, or if the delivery was deemed difficult by the practitioner. We considered that the impact of those events would be low after 3 months of age and set this variable to "0" for children older than 3 months.

We used a mixed logistic regression model to account for repeated measures (i.e., multiple consultations for the same child) with a random intercept for each child. Effects of exposures of primary interest on inappropriate antibiotic prescription were assumed to be the same for all children and, accordingly, were modeled as fixed effects. Upon recommendation from a statistical reviewer, we conducted a full model including data from all 3 countries. First, we conducted the full model without any interaction terms (S2 Table). Then, we tested possible interactions between each covariate and the country effect by considering each possible country:covariate interaction term one at a time (S3 Table). As 16 tests were performed, we applied Bonferroni's correction to counteract inflated type I error. The Bonferroni-corrected alphavalue considered for statistical significance of a variable was calculated as 0.05/16 = 0.003. In the final model, only the interaction terms identified as significant in the previous step were included.

A secondary analysis was also carried out, in which separate mixed logistic regression models were developed for each country to identify risk factors for inappropriate prescribing that are directly reflective of the local contexts (S4 and S5 Tables). For this analysis, both univariable and multivariable analyses were performed. For the multivariable analysis, all fixed-effect variables examined in univariable analysis were included and a backward stepwise Akaike information criterion (AIC) model selection procedure was conducted to select the variables resulting in the model with the lowest AIC. This approach was selected as it has interesting properties, namely that it is asymptotically efficient even if the model is misspecified [19]. All analyses were performed using R software version 4.1.2 (2021-11-01).

Ethics

The study was approved by the respective ethics committees of Madagascar (068-MSANP/CE), Senegal (SEN 14–20), Cambodia (108 NEHCR), and the Institutional Review Board of Institut Pasteur (IRB/2016/08/03), France. Written informed consent was given by parents of all participants.

Results

Study population and consultation characteristics

Among the 3,710 children enrolled at birth between 2012 and 2018, 2,719 (73%) children had at least 1 outpatient medical consultation during their follow-up and were included in this study, including 572 children in Cambodia, 1,816 children in Madagascar, and 331 in Senegal (Fig 1). Overall, 81.6% of included children had a complete follow-up in Cambodia, 94.0% (pilot phase) and 42.9% (complete phase) in Madagascar, and 59.8% (urban site) and 75.9% (rural site) in Senegal. The mean duration of follow-up was 670 days in Cambodia, 160 (pilot phase) and 372 (complete phase) days in Madagascar, and 258 (urban site) and 73 (rural site) days in Senegal.

The proportion of mothers who completed secondary school or attended university was 24.3% in Madagascar, compared to 13.6% in Cambodia and 9.1% in Senegal (p < 0.001,



Fig 1. Flow chart of children included in this study (yellow box), their outpatient consultations (green boxes), and their antibiotic prescriptions (blue boxes). The number of antibiotic prescriptions exceeds the number of consultations receiving prescriptions because some consultations resulted in multiple prescriptions.

https://doi.org/10.1371/journal.pmed.1004211.g001

Table 1). Most households had electricity in Cambodia (99.5%) and Senegal (98.2%), but just 75.6% of households in Madagascar (p < 0.001). Most mothers gave birth in a healthcare facility in Cambodia (98.6%) and Senegal (96.1%), while 37.7% of mothers delivered at home in Madagascar (p < 0.001).

Among the 2,719 children having at least 1 consultation, a total of 11,762 outpatient consultations occurred over the follow-up period. Among these, 3,448 (29.3%) consultations led to antibiotic prescriptions. These consultations represented 59.9% (1,409/2,352), 19.4% (1,733/ 8,916), and 61.9% (306/494) of all consultations in Cambodia, Madagascar, and Senegal, respectively (Fig 1).

In both Cambodia and Madagascar, the most frequent diagnosis was rhinopharyngitis, representing 34.6% (814/2,352) and 29.5% (2,634/8,916) of consultations, respectively. Gastroenteritis was the next most common diagnosis, accounting for 17.6% (416/2,352) of consultations in Cambodia and 12.5% (1,112/8,916) in Madagascar. The third most common diagnosis was bronchiolitis in Cambodia, with 8.9% (210/2,352) of consultations, and lower respiratory infection in Madagascar, with 10.8% (964/8,916) of consultations. In Senegal, the 3 most common diagnoses were bronchiolitis with 15.6% of consultations (77/494), healthy child (consultations without specific diagnoses/symptoms) with 13.3% of consultations (66/ 494), and rhinopharyngitis with 10.9% of consultations (54/494).

Across all consultations that led to antibiotic prescriptions, a total of 4,648 distinct antibiotics were prescribed, as some prescriptions included multiple antibiotics. The most prescribed antibiotic in Cambodia and Madagascar was amoxicillin, accounting for 35.7% (653/1,829) and 24.0% (597/2,488) of all antibiotics prescribed, respectively. Cefixime was the most prescribed antibiotic in Senegal, representing 34.4% (114/331) of all antibiotics prescribed (S6 Table).

Inappropriate antibiotic prescriptions

Of the 11,762 outpatient consultations, 10,416 were associated with diagnoses (88.6%) considered not to require antibiotic therapy. Of these, 25.3% (2,639/10,416, 95% CI [24.5, 26.2]) led to an antibiotic prescription. This proportion differed greatly among the 3 countries: 57.0% (1,174/2,060, 95% CI [54.8, 59.1]) in Cambodia, 15.6% (1,239/7,961, 95% CI [14.7, 16.4]) in

		Country			
Characteristics		Cambodia n (%)	Madagascar n (%)	Senegal n (%)	p-values ¹
N = 2,719		572	1,816	331	
Site	Rural	263 (46.0)	1,097 (60.4)	87 (26.3)	< 0.001
Sex	Male	280 (49.1)	935 (51.5)	166 (50.9)	0.476
	Missing data	2 (0.3)	0 (0.0)	5 (1.5)	
Mother's level of educatio	n				< 0.001
	None or primary school	314 (54.9)	436 (24.0)	249 (75.2)	
	Incomplete secondary	180 (31.5)	939 (51.7)	52 (15.7)	
	Secondary or university	78 (13.6)	441 (24.3)	30 (9.1)	
Mother's profession					< 0.001
	Executive or office job	23 (4.0)	102 (5.6)	6 (1.8)	
	Manual job	332 (58.0)	453 (25.0)	81 (24.5)	
	Student or unemployed	217 (37.9)	1,260 (69.4)	244 (73.7)	
	Missing data	0 (0.0)	1 (0.1)	0 (0.0)	
Mother's age (mean (SD))	[range]	27.39 (5.54) [15–46]	26.02 (6.5) [15-48]	28.49 (6.7) [15–46]	< 0.001
Parity	First child	241 (42.1)	673 (37.1)	77 (23.3)	< 0.001
	Missing data	0 (0.0)	1 (0.1)	0 (0.0)	
History of deceased child	Yes	19 (3.3)	130 (7.2)	17 (5.1)	0.001
	Missing data	0 (0.0)	1 (0.1)	0 (0.0)	
Latrine	Exterior	272 (47.6)	1,679 (92.5)	193 (58.7)	< 0.001
	Missing data	0 (0.0)	1 (0.1)	2 (0.6)	
Electricity	Yes	569 (99.5)	1,373 (75.6)	324 (98.2)	< 0.001
	Missing data	0 (0.0)	1 (0.1)	1 (0.3)	
Density of the house	Overcrowded	274 (47.9)	625 (34.4)	110 (33.2)	< 0.001
	Missing data	1 (0.2)	1 (0.1)	5 (1.5)	
Place of delivery	Health facility	563 (98.6)	1,132 (62.3)	317 (96.1)	< 0.001
	Missing data	1 (0.2)	0 (0.0)	1 (0.3)	
Cesarean section	Yes	71 (12.5)	180 (9.9)	11 (3.4)	< 0.001
	Missing data	2 (0.3)	0 (0.0)	8 (2.4)	
Resuscitation at birth					0.268
	No	518 (90.6)	1,628 (89.6)	292 (88.2)	
	Yes	49 (8.6)	183 (10.1)	21 (6.3)	
	Missing data	5 (0.8)	5 (0.3)	18 (5.5)	
Dystocic delivery					< 0.001
	No	524 (91.6)	1,625 (89.5)	278 (84.0)	
	Yes	33 (5.8)	178 (9.8)	6 (1.8)	
	Missing data	15 (2.6)	13 (0.7)	47 (14.2)	
Low birth weight*					0.002
	No	538 (94.1)	1,642 (90.4)	257 (77.6)	
	Yes	26 (4.5)	160 (8.8)	31 (9.4)	
	Missing data	8 (1.4)	14 (0.8)	43 (13.0)	
Premature delivery ^{\$}					0.123
	No	544 (95.1)	1,691 (93.1)	280 (84.6)	
	Yes	19 (3.3)	66 (3.6)	5 (1.5)	

Table 1. Description of the characteristics of the 2,719 children from the BIRDY cohort who had at least 1 outpatient medical consultation and were hence included in this study.

(Continued)

PLOS MEDICINE

Table 1. (Continued)

	Country			
Characteristics	Cambodia n (%)	Madagascar n (%)	Senegal n (%)	<i>p</i> -values ^µ
	9 (1.6)	59 (3.2)	46 (13.9)	

*Low birth weight is defined as birth weight <2,500 g.

^{\$}Birth before 37 weeks gestation.

^µp-values were calculated based on non-missing data and using Chi-square tests for categorical variables and F-tests for continuous variables.

https://doi.org/10.1371/journal.pmed.1004211.t001

Madagascar, and 57.2% (226/395, 95% CI [52.2, 62.2]) in Senegal (p < 0.001). Altogether, 76.5% (2,639/3,448, 95% CI [75.1, 77.9]) of all outpatient consultations resulting in antibiotic prescriptions did not require them, including 83.3% (1,174/1,409, 95% CI [81.3, 85.2]) in Cambodia, 71.5% (1,239/1,733, 95% CI [69.3, 73.6]) in Madagascar, and 73.9% (226/306, 95% CI [68.6, 78.7]) in Senegal (p < 0.001). If we consider distinct antibiotic prescriptions, altogether 67.1% (3,117/4,648, 95% CI [65.7, 68.4]) of all prescriptions were inappropriate, including 74.8% (1,368/1,829, 95% CI [72.7, 76.7]) in Cambodia, 60.8% (1,512/2,488, 95% CI [58.8, 62.7]) in Madagascar, and 71.6% (237/331, 95% CI [66.4, 76.4]) in Senegal (p < 0.001) (Fig 1).

Among the consultations determined not to require antibiotics, the diagnoses accounting for the greatest absolute share of inappropriate prescribing in Cambodia were rhinopharyngitis, with 480 of 814 (59.0%) associated consultations resulting in prescriptions and non-bloody gastroenteritis with 253 of 411 consultations (61.6%) (Fig 2). In Madagascar, these diagnoses also accounted for the greatest share of inappropriate prescribing, with antibiotic prescriptions resulting from 209 of 2,634 (7.9%) rhinopharyngitis consultations and 270 of 1,097 (24.6%) non-bloody gastroenteritis consultations. In Senegal, uncomplicated bronchiolitis accounted for the greatest absolute number of inappropriate prescriptions at 65 of 77 (84.4%) consultations. The proportion of consultations in which children were prescribed antibiotics despite



Fig 2. Number of consultations "not requiring antibiotics" stratified by country, associated diagnosis and the share resulting in antibiotic prescription (blue). Across all countries, N = 10,416. *Number of consultations with this diagnosis (percentage resulting in antibiotic prescription).

https://doi.org/10.1371/journal.pmed.1004211.g002



The 11 most common inappropriately prescribed antibiotic

Fig 3. Number of antibiotics prescribed for diagnoses that did not require antibiotics, stratified by country and type of antibiotic. N = 3,117 distinct antibiotics prescribed.

https://doi.org/10.1371/journal.pmed.1004211.g003

being diagnosed as in "good health" was 1.9% in Cambodia (1/53), 7.3% in Madagascar (40/ 545), and 78.8% in Senegal (52/66).

The most frequently prescribed inappropriate antibiotic was amoxicillin in Cambodia and Madagascar, accounting for 42.1% (576/1,368) and 29.2% (441/1,512) of inappropriate prescriptions, respectively. In Senegal, cefixime was the antibiotic most frequently prescribed inappropriately, representing 31.2% of inappropriate prescriptions (74/237) (Fig 3).

Determinants of inappropriate antibiotic prescribing

Table 2 describes determinants of antibiotic prescribing among consultations with a diagnosis determined not to require antibiotic therapy. Covariates with significant country-level effects (i.e., significant country:covariate interactions) were age and site (S3 Table).

We found that older age at consultation was associated with higher risk of receiving inappropriate antibiotic prescription in all 3 countries. Children aged >3 months had higher risk of inappropriate prescription compared to children aged <3 months, with adjusted odds ratios (aORs) for children aged 3 months to 1 year estimated at 1.91 (95% CI [1.63, 2.25]; p < 0.001), 3.34 (95% CI [1.95, 5.75]; *p* < 0.001), and 4.51 (95% CI [3.34, 6.08]; *p* < 0.001) in Madagascar, Senegal, and Cambodia, respectively. For children followed up beyond 1 year of age, aORs

		Madagascar, Cambodia, and Senegal n = 10,336 consultations among 2,509 children*	
Variables		Adjusted OR (95% CI)	P ¹
Age ²	<3 mo	ref	<0.001
	3 mo-1 yr	4.51 (3.34–6.08)	
	>1 yr	5.25 (3.85-7.15)	
Weight z-score	Normal	ref	0.670
	Underweight	1.03 (0.87–1.22)	
History of hospitalization in the las	st 90 days = Yes	0.94 (0.75–1.18)	0.628
History of antibiotic prescription in	n the last 15 days = Yes	1.50 (1.16–1.94)	0.002
Severity score	0	Ref	<0.001
	1	2.00 (1.75–2.30)	
	2	3.10 (2.47–3.91)	
Season = Rainy		1.32 (1.19–1.47)	<0.001
Complicated delivery = Yes		1.09 (0.85–1.39)	0.487
Country	Cambodia	ref	<0.001
	Madagascar	0.28 (0.20–0.38)	
	Senegal	2.60 (1.67–4.04)	
Site = Rural		4.07 (3.12–5.31)	<0.001
Sex = Male		1.08 (0.97–1.21)	0.181
Mother's level of education			
	None or primary school	ref	0.576
	Incomplete secondary	0.89 (0.78–1.02)	
	Secondary or university	0.89 (0.74–1.06)	
Mother's age	<26 yrs	1.10 (0.96–1.25)	0.168
Mother's profession	Manual	ref	0.570
	Executive or office job	0.90 (0.67–1.20)	
	Student or unemployed	0.94 (0.82–1.07)	
Parity	First child	0.94 (0.81–1.08)	0.393
History of deceased child	Yes	0.98 (0.78–1.24)	0.859
House density	Normal	ref	0.012
(Overcrowded (4 or more)	1.17 (1.03–1.32)	
Place of delivery	Health facility	ref	0.103
	At home	1.13 (0.98–1.31)	
Site x Country (Reference: for site: Cambodia	urban and for country:		<0.001
	Rural: Madagascar	0.45 (0.33–0.61)	
	Rural: Senegal	1.08 (0.54–2.14)	
Age x Country Reference: for age: < Cambodia	<3 mo and for country:		<0.001
	3 mo–1 yr: Madagascar	0.42 (0.31–0.58)	
3 mo-1 yr: Senegal		0.74 (0.40–1.36)	
>1 yr: Madagascar		0.69 (0.48–0.99)	
	>1 yr: Senegal	NA	
Country adj	usted ORs taking into acco	ount interaction parameters ³	
Site = Rural			
	Cambodia	4.07 (3.12–5.31)	
	Madagascar	1.83 (1.57–2.14)	

Table 2. Analysis of the determinants of antibiotic prescription among consultations of children of the 3 countries (Cambodia, Madagascar, and Senegal) with a diagnosis not requiring antibiotic therapy.

(Continued)

Table 2. (Continued)

	Madagascar, Cambodia, and Senegal n = 10,336 consultations among 2,509 children*		
Variables	Adjusted OR (95% CI)	P ¹	
Senegal	4.40 (2.34-8.28)		
Age ²			
3 mo-1 yr			
Cambodia	4.51 (3.34-6.08)		
Madagascar	1.91 (1.63–2.25)		
Senegal	3.34 (1.95–5.75)		
>1 yr			
Cambodia	5.25 (3.85-7.15)		
Madagascar	3.65 (2.96-4.48)		
Senegal	NA		

¹*P*-values are calculated from individual Wald tests. The threshold for *p*-values significance was set at 0.003 after Bonferroni correction.

²In Senegal, ORs associated with age at consultation between 3 months and 1 year should be interpreted only for infants of the urban site (no infants >3 months included in the rural site) and ORs associated with age at consultation >1 year should not be interpreted (no infants >1 year included).

³Adjusted ORs were calculated by considering both fixed and interaction effects for Madagascar and Senegal. Confidence intervals were obtained by considering both the variances of the estimates and the covariance between them.

*Analysis performed on consultations with complete dataset (for details, see <u>S5 Table</u>).

Ref, reference; OR, odds ratio; 95% CI, 95% confidence interval; NA, Not Applicable.

https://doi.org/10.1371/journal.pmed.1004211.t002

were 3.65 (95% CI [2.96, 4.28]; *p* < 0.001) and 5.25 (95% CI [3.85, 7.15]; *p* < 0.001) in Madagascar and Cambodia, respectively.

Consultations during the rainy season were associated with higher risk of inappropriate antibiotic prescription (aOR = 1.26, 95% CI [1.32, 1.47]; p < 0.001). Children living in rural areas had a higher risk of inappropriate prescription compared to those living in urban areas, with aORs of 1.83 (95% CI [1.57, 2.14]; p < 0.001) in Madagascar, 4.40 (95% CI [2.34, 8.28]; p < 0.001) in Senegal, and 4.07 (95% CI [3.12, 5.31]; p < 0.001) in Cambodia.

Having already received an antibiotic prescription in the 15 days prior to the consultation also increased the risk of inappropriate prescription (aOR = 1.50; 95% CI [1.16, 1.94]; p = 0.002). Finally, children with higher clinical severity score had a higher risk of inappropriate antibiotic prescription compared to children with a lower score, with aORs ranging from 2.00 (95% CI [1.75, 2.30]) to 3.10 (95% CI [2.47, 3.91]) for children with severity scores of = 1 or = 2, respectively, relative to severity score = 0.

In the secondary analysis considering a distinct model per country, results were highly similar. The 4 variables that were identified as relevant across each of the 3 countries (age at consultation, season, site, and severity score) were also identified as statistically significant in the primary analysis considering a single model (S4 Table).

Discussion

In this multicentric community cohort study conducted among children under 2 years old in Madagascar, Cambodia, and Senegal, we observed that approximately 1 quarter of outpatient consultations not requiring antibiotic therapy nonetheless resulted in an antibiotic prescription. Overall, approximately 3 quarters of consultations resulting in antibiotic prescription did not require them, and approximately two thirds of all antibiotics prescribed to these young

children were not necessary. These inappropriate antibiotics may contribute to the emergence of antibiotic resistance despite providing little to no benefit—or even harm—to the patients concerned.

Our results suggest that inappropriate antibiotic prescribing is frequent but highly heterogeneous across countries. In particular, approximately 57% of consultations not requiring antibiotics resulted in antibiotic prescriptions in Cambodia and Senegal, but only 16% in Madagascar. Although relatively few previous studies have estimated the share of inappropriate antibiotic prescribing in LMICs, substantial heterogeneity across countries is nonetheless consistent with the literature. In a secondary data analysis study, Sulis and colleagues estimated the proportion of patient–provider interactions resulting in inappropriate antibiotic prescriptions in 3 LMICs (India, China, and Kenya). They found that approximately 50% of interactions in India and Kenya resulted in inappropriate prescriptions, compared to approximately 29% in China [20]. However, this study considered both children and adults and only included certain preselected diagnoses preventing direct comparison with our study, which included all diagnoses among young children only. Our findings thus describe inappropriate antibiotic prescribing across all outpatient health events, resulting in a more complete and representative assessment of local prescribing practices. For this reason, diagnosis-specific estimates of inappropriate antibiotic prescribing may be more easily compared with the literature.

For children with gastroenteritis not requiring antibiotics, we found that the share of consultations resulting in antibiotic prescription was 24.6%, 51.2%, and 61.6% in Madagascar, Senegal, and Cambodia, respectively. In Senegal, our estimates can be compared to those of a study conducted in children under 5 with gastroenteritis in a primary healthcare in Dakar (the capital city of Senegal). In this study, in which all episodes were microbiologically documented, the authors found that 36% of children received antibiotics when no pathogens were detected in the stool [21]. In a study conducted in Madagascar's capital city, Antananarivo, approximately 47.2% of children with diarrhea received an antibiotic prescription. However, the authors did not distinguish between diarrheal episodes that required antibiotics and those that did not. In this study, physicians were found to justify antibiotic prescribing mainly by the presence of fever [22]. According to WHO, the prescription of antibiotics during an episode of diarrhea is indicated in the presence of bloody diarrhea or documented shigellosis. In such cases, diarrhea is frequently accompanied by fever, but fever is likely to accompany diarrhea regardless of bacterial or viral etiology [16]. One possible explanation for our results is that prescribers had similar practices, as fever was present in 47.8%, 41.1%, and 31.8% of gastroenteritis consultations resulting in an inappropriate prescription in Madagascar, Cambodia, and Senegal, respectively. In a separate study of children <2 years old consulting primary care in LMICs, Rogawski and colleagues estimated that 44.2% of non-bloody diarrhea consultations resulted in antibiotic prescribing, with estimates ranging from 9% to 57% across the 8 countries considered [23]. Finally, Sulis and colleagues estimated that the share of pediatric diarrhea consultations resulting in inappropriate antibiotic prescription ranged from 27.4% in China to 59.4% in India. Overall, these results support the extensive intercountry heterogeneity observed in our study.

For respiratory illness (upper and lower combined) not requiring antibiotics, Rogawski and colleagues further estimated that between 20.2% and 62.2% of consultations, depending on the country, resulted in inappropriate prescription [23]. By comparison, we found that approximately 1 quarter of uncomplicated lower respiratory infection (mainly bronchitis) consultations resulted in inappropriate prescriptions in Madagascar, but over 80% in Cambodia and Senegal. For rhinopharyngitis consultations, we found that 7.9%, 20.4%, and 59.0% resulted in inappropriate antibiotic prescription in Madagascar, Cambodia, and Senegal, respectively. These estimates are consistent with the extensive heterogeneity between countries observed by

Rogawski and colleagues, with very high inappropriate antibiotic prescribing for respiratory illness in certain countries [23]. This heterogeneity may be explained by various factors, including local care-seeking behaviors and prescribing practices, as well as variation in the underlying epidemiological burden of disease.

In Senegal, one striking finding is that consultations reported as "healthy" represented one of the most common diagnoses. Reviewing these diagnoses in detail, we found that all consultations reported as "healthy" took place at birth and occurred mainly in the rural site (87%). In addition, there was only 1 "healthy" consultation per child. This seems to be a pattern that is specific to Senegal, reflecting local consultation behavior or prescribing practices, and this result should therefore be interpreted in that specific context. Further studies should be conducted to better understand motivations of both mothers to consult for their newborn at delivery and physicians to prescribe antibiotics despite no clear indication.

Our study also provides information on all-cause antibiotic exposure (regardless of diagnosis or symptoms), which has been more frequently described in the literature. A recent metaanalysis evaluated all-cause antibiotic prescribing across all age groups in primary care in LMICs, estimating that 44% to 60% of consultations resulted in antibiotic prescriptions across 48 studies [12]. However, the authors highlight extensive heterogeneity between studies, which is mainly explained by differences in study design and makes comparison difficult. In our study, all-cause antibiotic prescribing ranged from approximately 20% of consultations in Madagascar to 60% of consultations in Cambodia and Senegal. Stark differences observed between Madagascar and the 2 other countries are not readily explained by different study designs (the protocol was identical for all 3 countries) nor by substantial differences in infectious disease burden (the types and distributions of diagnoses seem to be relatively similar).

By contrast, choice of antibiotic was more similar between Cambodia and Madagascar than Senegal. In particular, the most prescribed antibiotic was amoxicillin in Madagascar and Cambodia (24.0% and 35.7%, respectively), but cefixime (34.4%) in Senegal. Padget and colleagues conducted face-to-face interviews with caregivers of 1,401 children living in the same urban and rural sites in Madagascar and the same urban site in Senegal as in our study and examined antibiotic consumption in the community [24]. Although amoxicillin was the most used antibiotic in all study areas, the authors found that cephalosporins represented 15.9% of antibiotics consumed in the urban site in Senegal compared to 3.5% and 4.5% in the urban and rural sites in Madagascar, respectively. Our results are therefore consistent with these previous findings that cephalosporins (including cefixime) are more widely used in Senegal, possibly reflecting differences in prescribing habits of healthcare providers across these settings.

A study on the sale of antibiotics across low-, middle-, and high-income countries suggests that extensive intercountry variation in both the quantity and classes of antibiotics sold cannot be explained solely by differences in disease burden [25]. Rather, differences may in large part reflect heterogeneity in local healthcare practices, culture, and socioeconomic contexts. However, this analysis was based on drug sales data, thus providing limited information on prescribing practices.

Another recent spatial modeling study provided estimates of global antibiotic consumption based on data from Demographic Health Surveys and UNICEF's Multiple Indicator Cluster Surveys [26]. The authors have also shown wide variation in antibiotic consumption, both between and within countries. Lack of access to healthcare, particularly in poor and rural communities, as well as particular cultural and socioeconomic contexts might explain these disparities. However, as acknowledged by the authors, although both data sources used in this modeling study provide highly useful data in LMICs, they are subject to biases that may limit the interpretability of the results of this study. Using reliable, prospective, standardized cohort data that facilitates comparisons across sites and countries, our findings thus suggest disparities in antibiotic exposure among children across 3 LMICs.

In this study, we also identified leading determinants of inappropriate antibiotic prescribing. We found that children older than 3 months had a higher risk of receiving an inappropriate prescription than younger infants. This may be explained by a higher risk of severe bacterial infection during the postnatal period and the fact that several diagnoses that are more common in these youngest children require antibiotics [27-30]. This effect of age is consistent with findings from a study of children presenting with suspected pneumonia to a primary healthcare facility in Zambia, which found a higher proportion of incorrect antibiotic prescriptions for children aged 12 to 59 months than children under 12 months [31]. However, the authors did not identify any factors explaining this result.

Clinical severity score was also associated with a higher risk of inappropriate prescription. During consultations in which patients express more frequent or severe symptoms, parents or guardians may be more likely to demand antibiotics. During such consultations, practitioners may also choose to prescribe antibiotics due to concern about the risk of bacterial superinfection. These factors may be particularly relevant for certain diagnoses with extensive symptom expression and known risk of superinfection, such as bronchiolitis, which received a large share of inappropriate prescriptions in this study.

We found that consultations occurring in the rainy season were more likely to result in inappropriate prescriptions than those in the dry season. During rainy periods, respiratory and fecal-oral infections generally circulate more widely, which may have an impact on practitioners' prescribing practices. If the overall number of consultations for infectious pathologies increases, the practitioner's choice to prescribe an antibiotic during subsequent consultations may be influenced. Although increases in the total volume of antibiotic prescribing are well observed in winter months in HICs and have been documented in the rainy season in LMICs, seasonality in inappropriate antibiotic prescribing is less well documented, particularly in LMICs [32–34].

We found that children living in rural areas were at increased risk of receiving inappropriate antibiotic prescriptions. This may be explained by a limited presence of qualified health professionals in rural areas, especially physicians. Indeed, several studies have shown that rural areas suffer from a form of medical staff desertification, where public sector physicians initially work in rural areas after graduating from medical school but later move to urban areas [35,36]. Thus, in some instances, the medical staff that remains in these areas are individuals with backgrounds in healthcare but who are not qualified as physicians. Decreased access to healthcare facilities in rural areas may also explain this increased risk, particularly in the context of a practitioner's potential concern regarding bacterial superinfection, as patients in rural settings may be less able to re-consult in the days following an initial consultation. This is consistent with previous findings from Madagascar by Padget and colleagues, who reported that antibiotic prescriptions increase overall as the density of healthcare facilities decrease [24].

Our results suggest a need for interventions to prevent the inappropriate prescribing of antibiotics to children in LMICs. One possible intervention is to train prescribers with validated pediatric clinical management guidelines. For example, training practitioners with IMCI guidelines has been shown previously to decrease inappropriate antibiotic prescribing in Tanzania and Uganda [37]. Also, lack of access to microbiological documentation in LMICs may explain frequent antibiotic prescription in case of doubt of bacterial origin of the disease. Rapid diagnostic testing may also help prescribers, especially those lacking training or practicing in remote areas, to understand whether a disease is likely to be of viral rather than bacterial etiology. For example, rapid tests are widely used in HICs to identify sore throat due to group A *Streptococcus* and may help to prevent inappropriate antibiotic prescribing by suggesting when viral origin may be more likely [38]. For bronchiolitis, rapid tests to identify respiratory syncytial virus have also demonstrated good specificity, although their use in routine practice remains to be evaluated [39]. C-reactive-protein point-of-care tests may also prove beneficial to improve antibiotic prescription, although optimal cut-off thresholds to distinguish between bacterial and nonbacterial infection across diverse use cases remain to be determined [40,41].

To our knowledge, our study is the first in LMICs to estimate the proportion of outpatient consultations resulting in inappropriate antibiotic prescriptions in young children. It is also the first to estimate the total share of antibiotic prescriptions that are inappropriate in these settings. Our prospective, longitudinal study design with rigorous documentation of patient symptoms and diagnoses facilitated the evaluation of antibiotic necessity for each consultation. This makes our findings a good reflection of local practices regarding both inappropriate and overall antibiotic prescribing, particularly as over-the-counter antibiotics have been shown to represent only a small portion of antibiotic consumption in children [24].

Our study also has several limitations. Bacteriological investigations were performed only in prespecified cases, therefore, not all infectious diagnoses were microbiologically confirmed, and in particular, those presumed to be of viral origin [14]. This may have resulted in misclassification of some diagnoses, including both those that we determined as requiring and not requiring antibiotics. We are unable to estimate the magnitude of these misclassifications, but they may have resulted in overestimation of inappropriate prescribing, given that the proportion of diagnoses not requiring antibiotics is much greater than those not (88.6% versus 11.4%). However, the study protocol provided access to laboratory tests as needed for all children included, at the clinician's discretion, to guide prescription [14]. We are also unable to assess how representative our results are of antibiotic prescribing in these settings in nonstudy conditions, in particular, because all monetary costs related to consultations and antibiotic prescriptions were supported by the project. Given this economic context, it is unlikely that children consulted external practitioners during the study period, suggesting that consultation histories should be relatively complete, although this may possibly have resulted in children consulting more frequently than they would normally in non-study conditions. As there were also no financial barriers to prescribing, it is possible that clinicians in our study prescribed antibiotics more frequently than they would in non-study conditions. Further, antibiotic access is known to influence antibiotic use, and reduced barriers to antibiotic access among our study participants may have led to increased probability of prescribed antibiotics being dispensed via formal channels (i.e., pharmacies) and thus captured by our study, and decreased probability of patients accessing informal or over-the-counter antibiotics [42-44].

Depending on the availability of particular healthcare providers at the time of the health event, as well as potential parental preferences, children could consult healthcare providers from either the public or private sectors. This could potentially bias findings as private sector providers may receive economic incentives to prescribe certain types of antibiotics [20]. Finally, our results cannot be generalized to the entire population because our study only concerned children under 2 years of age and included limited sites per country. In particular, the relatively small sample size in Senegal, which was limited due to logistical issues, may limit interpretation of results from this country. Despite these limitations, and in the context of considerable challenges to quantifying and evaluating the appropriateness of community antibiotic prescribing in LMICs, if interpreted with care, we believe that our study has facilitated reliable estimation of trends of inappropriate antibiotic prescribing among young children in these 3 LMICs.

In conclusion, in a community-based cohort of young children spanning 3 LMICs, we observed that the majority of all outpatient antibiotic prescriptions were inappropriate. While the proportion of consultations resulting in inappropriate prescriptions was variable across

countries, several common risk factors for inappropriate prescribing were nonetheless identified. These findings support the expansion of interventions to guide antibiotic prescribing that have already proven effective in certain LMIC settings, including training with validated clinical practice guidelines (e.g., IMCI) and the use of rapid diagnostic tests. The decision-making process underlying antibiotic prescription is complex, particularly in the absence of robust microbiological information and given challenging socioeconomic contexts, requiring the implementation of locally adapted multimodal strategies [45-47]. Future research is needed to better understand practitioners' motivations for prescribing and why they may deviate from established guidelines. Sociological studies and qualitative methods such as practitioner and patient interviews may facilitate a better understanding of antibiotic prescribing and its perceived necessity among high-risk populations in LMICs.

Supporting information

S1 Checklist. Strengthening the Reporting of Observational Studies in Epidemiology (STROBE) checklist.

(DOCX)

S1 Table. List of diagnoses defined as "not requiring antibiotic therapy," including diagnoses classified by the two-step algorithm to determine which health events do not require antibiotics in the absence of discriminating criteria. (DOCX)

S2 Table. Analysis of the determinants of antibiotic prescription among consultations of children of the 3 countries (Cambodia, Madagascar, and Senegal) with a diagnosis not requiring antibiotic therapy without any covariate: country interaction terms. (DOCX)

S3 Table. Results (χ 2, degrees of freedom, *p*-values) of including covariate:country interaction terms in the regression model, where each interaction term is considered one at a time.

(DOCX)

S4 Table. Analysis of the determinants of antibiotic prescription among consultations with a diagnosis not requiring antibiotic therapy, with a separate model considered for each country.

(DOCX)

S5 Table. Characteristics of consultations determined not to require antibiotic therapy. (DOCX)

S6 Table. Distribution of antibiotics prescribed across all consultations, stratified by country. N = 4,648 antibiotic prescriptions across all countries. (DOCX)

Acknowledgments

We are grateful to all the mothers and their newborns for their participation. We thank all physicians, laboratory staff, field interviewers, and community workers for their involvement in this project. We thank all collaborators of the BIRDY program: Rasoanaivo Fanjalalaina Vololonirina, Andrianonimiadana Lova Maminirina, Rabearitiana Lydos, Rasoloson Dimitri, Ratsima Hariniaina Elisoa, Volahasina Antsa Tanjona, Volahasina Antso Fenitra, Randriamamonjiarison Aina Nirina, Andriatahina Todisoa, Marguerite Diatta, Joseph Faye, Armya Youssouf Abdou, Sophie Goyet, Véronique Ngo, Frederique Randrianirina, Abdoulaye Seck, Elisoa Hariniaina-Ratsima, Michael Padget, and Benoit Garin. We also thank the Department of International Affairs and the Centre de Recherche Translationnelle—Coordination Clinique of Institut Pasteur for their support in the management of the study.

Author Contributions

Conceptualization: Elsa Kermorvant-Duchemin, Bich-Tram Huynh.

Data curation: Lison Ramblière, Bich-Tram Huynh.

Formal analysis: Antoine Ardillon, Lison Ramblière, Bich-Tram Huynh.

Funding acquisition: Elisabeth Delarocque-Astagneau, Didier Guillemot, Bich-Tram Huynh.

- **Investigation:** Touch Sok, Andrianirina Zafitsara Zo, Jean-Baptiste Diouf, Pring Long, Siyin Lach, Fatoumata Diene Sarr, Laurence Borand, Jean-Marc Collard, Perlinot Herindrainy, Agathe de Lauzanne, Muriel Vray.
- Methodology: Antoine Ardillon, Lison Ramblière, Elsa Kermorvant-Duchemin, Felix Cheysson, Bich-Tram Huynh.
- **Project administration:** Jean-Marc Collard, Perlinot Herindrainy, Agathe de Lauzanne, Muriel Vray.
- Supervision: Lison Ramblière, Fatoumata Diene Sarr, Laurence Borand, Jean-Marc Collard, Perlinot Herindrainy, Agathe de Lauzanne, Muriel Vray, Elisabeth Delarocque-Astagneau, Didier Guillemot, Bich-Tram Huynh.
- Validation: Elsa Kermorvant-Duchemin, Bich-Tram Huynh.

Visualization: Antoine Ardillon, Lison Ramblière.

- Writing original draft: Antoine Ardillon, Bich-Tram Huynh.
- Writing review & editing: Antoine Ardillon, Lison Ramblière, Elsa Kermorvant-Duchemin, Touch Sok, Andrianirina Zafitsara Zo, Jean-Baptiste Diouf, Pring Long, Siyin Lach, Fatoumata Diene Sarr, Laurence Borand, Felix Cheysson, Jean-Marc Collard, Perlinot Herindrainy, Agathe de Lauzanne, Muriel Vray, Elisabeth Delarocque-Astagneau, Didier Guillemot, Bich-Tram Huynh.

References

- Laxminarayan R, Matsoso P, Pant S, Brower C, Røttingen J-A, Klugman K, et al. Access to effective antimicrobials: a worldwide challenge. Lancet. 2016; 387:168–175. https://doi.org/10.1016/S0140-6736 (15)00474-2 PMID: 26603918
- Klein EY, Tseng KK, Pant S, Laxminarayan R. Tracking global trends in the effectiveness of antibiotic therapy using the Drug Resistance Index. BMJ Glob Health. 2019; 4:e001315. https://doi.org/10.1136/ bmjgh-2018-001315 PMID: 31139449
- Laxminarayan R, Duse A, Wattal C, Zaidi AKM, Wertheim HFL, Sumpradit N, et al. Antibiotic resistance —the need for global solutions. Lancet Infect Dis. 2013; 13:1057–1098. https://doi.org/10.1016/S1473-3099(13)70318-9 PMID: 24252483
- Klein EY, Van Boeckel TP, Martinez EM, Pant S, Gandra S, Levin SA, et al. Global increase and geographic convergence in antibiotic consumption between 2000 and 2015. Proc Natl Acad Sci U S A. 2018; 115:E3463–E3470. https://doi.org/10.1073/pnas.1717295115 PMID: 29581252
- Francino MP. Antibiotics and the Human Gut Microbiome: Dysbioses and Accumulation of Resistances. Front Microbiol. 2016; 6. Available from: https://www.frontiersin.org/articles/10.3389/fmicb.2015.01543. https://doi.org/10.3389/fmicb.2015.01543 PMID: 26793178
- Summers AO. Generally overlooked fundamentals of bacterial genetics and ecology. Clin Infect Dis. 2002; 34(Suppl 3):S85–S92. https://doi.org/10.1086/340245 PMID: 11988878

- Lipsitch M, Bergstrom CT, Levin BR. The epidemiology of antibiotic resistance in hospitals: paradoxes and prescriptions. Proc Natl Acad Sci U S A. 2000; 97:1938–1943. https://doi.org/10.1073/pnas.97.4. 1938 PMID: 10677558
- Fleming-Dutra KE, Hersh AL, Shapiro DJ, Bartoces M, Enns EA, File TM Jr, et al. Prevalence of Inappropriate Antibiotic Prescriptions Among US Ambulatory Care Visits, 2010–2011. JAMA. 2016; 315:1864–1873. https://doi.org/10.1001/jama.2016.4151 PMID: 27139059
- Pouwels KB, Hopkins S, Llewelyn MJ, Walker AS, McNulty CA, Robotham JV. Duration of antibiotic treatment for common infections in English primary care: cross sectional analysis and comparison with guidelines. BMJ. 2019; 364:I440. https://doi.org/10.1136/bmj.I440 PMID: 30814052
- Papenburg J, Fontela PS, Freitas RR, Burstein B. Inappropriate Antibiotic Prescribing for Acute Bronchiolitis in US Emergency Departments, 2007–2015. J Pediatric Infect Dis Soc. 2019; 8:567–570. https://doi.org/10.1093/jpids/piy131 PMID: 30657968
- Smieszek T, Pouwels KB, Dolk FCK, Smith DRM, Hopkins S, Sharland M, et al. Potential for reducing inappropriate antibiotic prescribing in English primary care. J Antimicrob Chemother. 2018; 73:ii36–ii43. https://doi.org/10.1093/jac/dkx500 PMID: 29490058
- Sulis G, Adam P, Nafade V, Gore G, Daniels B, Daftary A, et al. Antibiotic prescription practices in primary care in low- and middle-income countries: A systematic review and meta-analysis. PLoS Med. 2020; 17:e1003139. https://doi.org/10.1371/journal.pmed.1003139 PMID: 32544153
- Fink G D'Acremont V, Leslie HH, Cohen J. Antibiotic exposure among children younger than 5 years in low-income and middle-income countries: a cross-sectional study of nationally representative facilitybased and household-based surveys. Lancet Infect Dis. 2020; 20:179–187. https://doi.org/10.1016/ S1473-3099(19)30572-9 PMID: 31843383
- Huynh B-T, Kermorvant-Duchemin E, Chheang R, Randrianirina F, Seck A, Ratsima EH, et al. Severe bacterial neonatal infections in Madagascar, Senegal, and Cambodia: A multicentric community-based cohort study. PLoS Med. 2021; 18:e1003681. https://doi.org/10.1371/journal.pmed.1003681 PMID: 34582450
- Rambliere L, Kermorvant-Duchemin E, de Lauzanne A, Collard J-M, Herindrainy P, Vray M, et al. Excess risk of subsequent infection in hospitalized children from a community cohort study in Cambodia and Madagascar. Int J Epidemiol. 2022; 51:1421–1431. <u>https://doi.org/10.1093/ije/dyac048</u> PMID: 35333344
- Integrated management of childhood illness—Chart booklet (March 2014). [cited 2022 Jul 28]. Available from: https://www.who.int/publications/m/item/integrated-management-of-childhood-illness—chartbooklet-(march-2014).
- Bourrillon A, Benoist G, Chabrol B, Chéron G, Grimprel E. Pédiatrie pour le praticien—médecine de l'enfant et de l'adolescent. Elsevier Masson; 2020.
- World Health Organization. Child growth standards—Weight-for-age. [cited 2022 Jul 28]. Available from: https://www.who.int/tools/child-growth-standards/standards/weight-for-age.
- Ding J, Tarokh V, Yang Y. Model Selection Techniques—An Overview. IEEE Signal Process Mag. 2018; 35:16–34. https://doi.org/10.1109/MSP.2018.2867638
- 20. Sulis G, Daniels B, Kwan A, Gandra S, Daftary A, Das J, et al. Antibiotic overuse in the primary health care setting: a secondary data analysis of standardised patient studies from India, China and Kenya. BMJ Glob Health. 2020; 5:e003393. https://doi.org/10.1136/bmjgh-2020-003393 PMID: 32938614
- Sire J-M, Garin B, Chartier L, Fall NK, Tall A, Seck A, et al. Community-acquired infectious diarrhoea in children under 5 years of age in Dakar, Senegal. Paediatr Int Child Health. 2013; 33:139–144. https:// doi.org/10.1179/2046905512Y.000000046 PMID: 23930725
- Ravelomanana L, Tsifiregna RL, Rakotomalala LH, Razafimahatombo C, Ravelomanana N. Prescription of medications for acute diarrhea in infants in Antananarivo, Madagascar. Med Sante Trop. 2018; 28:182–185. https://doi.org/10.1684/mst.2018.0776 PMID: 29997077
- Rogawski ET, Platts-Mills JA, Seidman JC, John S, Mahfuz M, Ulak M, et al. Use of antibiotics in children younger than two years in eight countries: a prospective cohort study. Bull World Health Organ. 2017; 95:49–61. https://doi.org/10.2471/BLT.16.176123 PMID: 28053364
- 24. Padget M, Tamarelle J, Herindrainy P, Ndir A, Diene Sarr F, Richard V, et al. A community survey of antibiotic consumption among children in Madagascar and Senegal: the importance of healthcare access and care quality. J Antimicrob Chemother. 2017; 72:564–573. <u>https://doi.org/10.1093/jac/dkw446 PMID: 28115503</u>
- Hsia Y, Sharland M, Jackson C, Wong ICK, Magrini N, Bielicki JA. Consumption of oral antibiotic formulations for young children according to the WHO Access, Watch, Reserve (AWaRe) antibiotic groups: an analysis of sales data from 70 middle-income and high-income countries. Lancet Infect Dis. 2019; 19:67–75. https://doi.org/10.1016/S1473-3099(18)30547-4 PMID: 30522834

- 26. Browne AJ, Chipeta MG, Haines-Woodhouse G, Kumaran EPA, Hamadani BHK, Zaraa S, et al. Global antibiotic consumption and usage in humans, 2000–18: a spatial modelling study. Lancet Planet Health. 2021; 5:e893–e904. https://doi.org/10.1016/S2542-5196(21)00280-1 PMID: 34774223
- Craig JC, Williams GJ, Jones M, Codarini M, Macaskill P, Hayen A, et al. The accuracy of clinical symptoms and signs for the diagnosis of serious bacterial infection in young febrile children: prospective cohort study of 15 781 febrile illnesses. BMJ. 2010; 340:c1594. <u>https://doi.org/10.1136/bmj.c1594</u> PMID: 20406860
- Gomez B, Mintegi S, Bressan S, Da Dalt L, Gervaix A, Lacroix L, et al. Validation of the "Step-by-Step" Approach in the Management of Young Febrile Infants. Pediatrics. 2016; 138:e20154381. <u>https://doi.org/10.1542/peds.2015-4381</u> PMID: 27382134
- Cioffredi L-A, Jhaveri R. Evaluation and Management of Febrile Children: A Review. JAMA Pediatr. 2016; 170:794–800. https://doi.org/10.1001/jamapediatrics.2016.0596 PMID: 27322346
- Van den Bruel A, Haj-Hassan T, Thompson M, Buntinx F, Mant D. Diagnostic value of clinical features at presentation to identify serious infection in children in developed countries: a systematic review. Lancet. 2010; 375:834–845. https://doi.org/10.1016/S0140-6736(09)62000-6 PMID: 20132979
- Graham K, Sinyangwe C, Nicholas S, King R, Mukupa S, Källander K, et al. Rational use of antibiotics by community health workers and caregivers for children with suspected pneumonia in Zambia: a cross-sectional mixed methods study. BMC Public Health. 2016; 16:897. https://doi.org/10.1186/ s12889-016-3541-8 PMID: 27567604
- Suda KJ, Hicks LA, Roberts RM, Hunkler RJ, Taylor TH. Trends and Seasonal Variation in Outpatient Antibiotic Prescription Rates in the United States, 2006 to 2010. Antimicrob Agents Chemother. 2014; 58:2763–2766. https://doi.org/10.1128/AAC.02239-13 PMID: 24590486
- Elseviers MM. Ferech M, Vander Stichele RH, Goossens H, Group the E project. Antibiotic use in ambulatory care in Europe (ESAC data 1997–2002): trends, regional differences and seasonal fluctuations. Pharmacoepidemiol Drug Saf. 2007; 16:115–123. https://doi.org/10.1002/pds.1244 PMID: 16700079
- Van Boeckel TP, Gandra S, Ashok A, Caudron Q, Grenfell BT, Levin SA, et al. Global antibiotic consumption 2000 to 2010: an analysis of national pharmaceutical sales data. Lancet Infect Dis. 2014; 14:742–750. https://doi.org/10.1016/S1473-3099(14)70780-7 PMID: 25022435
- Zurn P, Dal Poz MR, Stilwell B, Adams O. Imbalance in the health workforce. Hum Resour Health. 2004; 2:13. https://doi.org/10.1186/1478-4491-2-13 PMID: 15377382
- Lassi ZS, Musavi NB, Maliqi B, Mansoor N, de Francisco A, Toure K, et al. Systematic review on human resources for health interventions to improve maternal health outcomes: evidence from low- and middle-income countries. Hum Resour Health. 2016; 14:10. <u>https://doi.org/10.1186/s12960-016-0106-y</u> PMID: 26971317
- 37. Gouws E, Bryce J, Habicht J-P, Amaral J, Pariyo G, Schellenberg JA, et al. Improving antimicrobial use among health workers in first-level facilities: results from the multi-country evaluation of the Integrated Management of Childhood Illness strategy. Bull World Health Organ. 2004; 82:509–515. PMID: 15508195
- van Houten CB, Naaktgeboren C, Buiteman BJM, van der Lee M, Klein A, Srugo I, et al. Antibiotic Overuse in Children with Respiratory Syncytial Virus Lower Respiratory Tract Infection. Pediatr Infect Dis J. 2018; 37:1077–1081. https://doi.org/10.1097/INF.00000000001981 PMID: 29601448
- Jung BK, Choi SH, Lee JH, Lee J, Lim CS. Performance evaluation of four rapid antigen tests for the detection of Respiratory syncytial virus. J Med Virol. 2016; 88:1720–1724. https://doi.org/10.1002/jmv. 24522 PMID: 26990654
- Escadafal C, Incardona S, Fernandez-Carballo BL, Dittrich S. The good and the bad: using C reactive protein to distinguish bacterial from non-bacterial infection among febrile patients in low-resource settings. BMJ Glob Health. 2020; 5:e002396. https://doi.org/10.1136/bmjgh-2020-002396 PMID: 32467355
- Althaus T, Greer RC, Swe MMM, Cohen J, Tun NN, Heaton J, et al. Effect of point-of-care C-reactive protein testing on antibiotic prescription in febrile patients attending primary care in Thailand and Myanmar: an open-label, randomised, controlled trial. Lancet Glob Health. 2019; 7:e119–e131. https://doi. org/10.1016/S2214-109X(18)30444-3 PMID: 30554748
- 42. Allwell-Brown G, Hussain-Alkhateeb L, Sewe MO, Kitutu FE, Strömdahl S, Mårtensson A, et al. Determinants of trends in reported antibiotic use among sick children under five years of age across low-income and middle-income countries in 2005–17: A systematic analysis of user characteristics based on 132 national surveys from 73 countries. Int J Infect Dis. 2021; 108:473–482. <u>https://doi.org/10.1016/j.ijid.2021.05.058</u> PMID: 34058373
- Gautham M, Miller R, Rego S, Goodman C. Availability, Prices and Affordability of Antibiotics Stocked by Informal Providers in Rural India: A Cross-Sectional Survey. Antibiotics. 2022; 11:523. <u>https://doi.org/10.3390/antibiotics11040523</u> PMID: 35453278

- 44. Do NTT, Vu HTL, Nguyen CTK, Punpuing S, Khan WA, Gyapong M, et al. Community-based antibiotic access and use in six low-income and middle-income countries: a mixed-method approach. Lancet Glob Health. 2021; 9:e610–e619. https://doi.org/10.1016/S2214-109X(21)00024-3 PMID: 33713630
- 45. Limato R, Nelwan EJ, Mudia M, Alamanda M, Manurung ER, Mauleti IY, et al. Perceptions, views and practices regarding antibiotic prescribing and stewardship among hospital physicians in Jakarta, Indonesia: a questionnaire-based survey. BMJ Open. 2022; 12:e054768. <u>https://doi.org/10.1136/bmjopen-2021-054768 PMID: 35589350</u>
- 46. Wilkinson A, Ebata A, MacGregor H. Interventions to Reduce Antibiotic Prescribing in LMICs: A Scoping Review of Evidence from Human and Animal Health Systems. Antibiotics (Basel). 2018; 8:2. <u>https://doi.org/10.3390/antibiotics8010002</u> PMID: 30583566
- Das J, Woskie L, Rajbhandari R, Abbasi K, Jha A. Rethinking assumptions about delivery of healthcare: implications for universal health coverage. BMJ. 2018; 361:k1716. https://doi.org/10.1136/bmj.k1716 PMID: 29784870